
Status, October 3
Apparently, the only problem today is that the soil mote at FLR has quit.  Is there a spare in the crater?

Also P.rim

ToDo:

Clean kh2o & radiometers at FLR (done yesterday at FAR and NEAR).

Deploy two sonics on the SW slope in the crater.

Shoot sonic azimuths.

Raise radiation stand at NEAR (done yesterday at FAR).
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